Author's response to reviews

Title: Pulmonary Function in Patients with Huntington's Disease

Authors:

    Alvaro Reyes (a.reyes@ecu.edu.au)
    Travis Cruickshank (t.cruickshank@ecu.edu.au)
    Mel Ziman (m.ziman@ecu.edu.au)
    Kazunori Nosaka (k.nosaka@ecu.edu.au)

Version: 2 Date: 20 January 2014

Author's response to reviews:

The email address of all authors of the manuscript has been added in the title page.
Acknowledgments have been included in the manuscript.